Author's response to reviews

Title: The height-, weight-, and BMI-for-age of Polish school-aged children and adolescents relative to international and local growth references

Authors:

Zbigniew Kulaga ZK (z.kulaga@czd.pl)
Mieczysław Litwin ML (m.litwin@czd.pl)
Marcin Tkaczyk MT (mtkaczyk@uni.lodz.pl)
Agnieszka Różdżyńska AR (a.rozdzynska@czd.pl)
Katarzyna Barwicka KB (kbarwicka@gmail.com)
Aneta Grajda AG (a.grajda@czd.pl)
Anna Świąder AS (a.swiader@czd.pl)
Beata Gurzkowska BG (b.gurzkowska@czd.pl)
Ewelina Napieralska EN (e.napieralska@czd.pl)
Huiqi Pan HP (h.pan@ich.ucl.ac.uk)

Version: 4 Date: 12 February 2010

Author's response to reviews:

Warszawa, 2010-02-12
Zbigniew Kulaga MD
Public Health Division
The Children’s Memorial Health Institute
Al. Dzieci Polskich 20
04-730 Warsaw
Poland

MS: 5835644182966668
The height-, weight-, and BMI-for-age of Polish school-aged children and adolescents relative to international and local growth refer
Zbigniew Kulaga ZK, Mieczysław Litwin ML, Marcin Tkaczyk MT, Agnieszka Różdżyńska AR, Katarzyna Barwicka KB, Aneta Grajda AG, Anna Świąder AS, Beata Gurzkowska BG, Ewelina Napieralska EN, Huiqi Pan HP

Dear Editor-in-Chief,

Thank you very much for accepting revised manuscript for publication in BMC Public Health.

On behalf of authors of the paper I submit the manuscript, in which the English used has been improved as recommended.
Yours faithfully,

Zbigniew Ku#aga